Author's response to reviews

Title: Watch Me Grow: A garden-based pilot intervention to increase fruit and vegetable intake in preschoolers

Authors:

Rebecca J N Brouwer (rebecca.brouwer@duke.edu)
Sara B Neelon (sara.neelon@duke.edu)

Version: 2 Date: 13 December 2012

Author's response to reviews: see over
Omar Khan, MD
Editor, BMC Public Health, BioMed Central
236 Gray's Inn Road
London WC1X 8HB, United Kingdom

December 13, 2012

Dear Dr. Khan,

Please accept the changes to our submission for the Research Article entitled “Watch Me Grow: A garden-based pilot intervention to increase fruit and vegetable intake in preschoolers.”

In this resubmission we have addressed the following changes:

- We provided additional context information in the Background Section of the Abstract.
- We inserted the title page with the requested information.
- We have provided information regarding the institutional review board.
- We complete inserted sections for competing interests and authors’ contributions
- We have added the trial registration information (ISRCTN34755197)

Sincerely,

Sara Benjamin Neelon, PhD, MPH, RD
Assistant Professor
Department of Community and Family Medicine
Duke University Medical Center and Duke Global Health Institute
sara.neelon@duke.edu